Author's response to reviews

Title: Severe anemia due to pharyngeal leech infestation; A case report from Ethiopia

Authors:

Nebiyu Shitaye (handworklove@gmail.com)

segenet shibabaw (segenetlove@gmail.com)

Version: 2 Date: 25 Jul 2017

Author’s response to reviews:

To reviewer 2:

Thank you for pushing me to do my best. I include the importance of sevoflurane during removal of leech at the last paragraph of discussion part as follows

...........The best way to remove the leech from the aerodigestive tract is direct laryngoscopic examination under general anesthesia. For induction sevoflurane-oxygen mixture via facemask is used. The leech is paralyzed and detached from its attachment by effect of sevoflurane............